Author’s response to reviews

Title: Nutrient Intakes and Sources of Fiber Among Children with Low and High Dietary Fiber Intake: The 2016 Feeding Infants and Toddlers Study (FITS), a Cross-sectional Survey

Authors:

Kristen Finn (kristen.finn@us.nestle.com)
Emma Jacquier (emma.jacquier@rd.nestle.com)
Brian Kineman (brian.kineman@us.nestle.com)
Heidi Storm (heidi.storm@us.nestle.com)
Ryan Carvalho (ryan.carvalho@nestle.com)

Version: 3 Date: 04 Nov 2019

Author’s response to reviews:

Dear Editor,

Thank you for your review of our revised manuscript and additional comments. Below we have provided detailed responses to the requested revisions.

1 In your Ethics approval section, please clarify whether consent obtained from participants was written or verbal. If consent was obtained verbally please include reasoning for this and state if an IRB approved this method of obtaining consent.

Furthermore, please detail the age of the caregivers of children to clarify whether or not they were of age to consent for their participation and any underage participants they may have proxied for.

We clarified these points in the ethics section.

2 Submission of a manuscript to a BioMed Central journal implies that materials described in the manuscript, including all relevant raw data, will be freely available to any scientist wishing to use them for non-commercial purposes, without breaching participant confidentiality. Thus, we request that you upload English language versions of all questionnaires developed specifically for use in this study as supplementary files. If the questionnaires are not yours or are already published by you, please supply references and/or links to them.
We provided a reference where details about the questionnaire are provided.

3 In your Funding section, please also state the role of the funding body in the design of the study; collection, analysis, and interpretation of data; and in writing the manuscript.

Clarified this in the funding section.

4 At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Clean manuscript provided.

Also, we also request that Ryan Carvalho be indicated as the corresponding author on the final publication.